Author's response to reviews

Title: Associations of GC and NADSYN1 polymorphisms with vitamin D and metabolic profile in Non-diabetic adults

Authors:

Lydia FOUCAN (lfoucan@yahoo.fr)
Fritz-Line VELAYOUNDOM-CEPHISE (fritz-line.cephise@chu-guadeloupe.fr)
Laurent LARIFLA (laurent.larifla@chu-guadeloupe.fr)
Christophe ARMAND (christophe.armand@chu-guadeloupe.fr)
Jacqueline DELOUMEAUX (jacqueline.deloumeaux@chu-guadeloupe.fr)
Cédric FAGOUR (Cedric.Fagour@chu-fortdefrance.fr)
Jean PLUMASSEAU (france.plumasseau@orange.fr)
Marilyn PORT-LIS (marilyn.port-lis@chu-guadeloupe.fr)
Longjian LIU (ll85@drexel.edu)
Fabrice BONNET (Fabrice.BONNET@chu-rennes.fr)
Jacques DUCROS (ducrosjacques@hotmail.fr)

Version: 3 Date: 20 September 2013

Author's response to reviews:

Responses to the Associate Editor comments.

Title: Associations of GC and NADSYN1 polymorphisms with vitamin D and metabolic profile in healthy adults

I - "References appear twice in this manuscript.

R – The second references list has been deleted.

II - In Table 1, for categorical variables the parenthesis next to the variable name indicate that data presented in the subsequent columns are percentages when this is, in fact, not the case.

The columns represent numbers (percentages). As an example, the first column for variable Overweight currently reads Overweight (%); it should be: Overweight (n, %). Similar corrections are needed for Obesity, Abdominal obesity, Hypertension, Dyslipidemia, and Vitamin D Insufficiency."

R : The corrections were made for all the categorical variables in table 1 as (n (%)).